Author's response to reviews

Title: Clinical and histogenic features of botryoid odontogenic cyst

Authors:

Vitor H Farina (odontofarina@uol.com.br)
Adriana AH Brandão (aigotti@fisjc.unesp.br)
Luiz AG Cabral (cabral@fisjc.unesp.br)
Janete D Almeida (janete@fisjc.unesp.br)

Version: 3 Date: 5 April 2010

Author's response to reviews: see over
April, 2010.

We are sending the revised article “CLINICAL AND HISTOLOGICAL FEATURES OF BOTRYOID ODONTOGENIC CYST: A CASE REPORT” to the Journal of Medical Case Reports for evaluation and possible publication.

As requested, this version was revised and includes a final conclusion (in red).

We appreciate your consideration in this article and we will be waiting an answer as soon as possible.

Yours sincerely,

Janete Dias Almeida, PhD
Department of Biosciences and Diagnosis
São José dos Campos Dental School
São Paulo State University – UNESP
Brazil
e-mail: janete@fosjc.unesp.br